Reviewer 1:
Really interesting and well written article.

Reply:
We are very grateful for the encouragement given by the reviewers and we will further improve ourselves in our future work.

Science Editor:

(1) Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. If the author has used a figure published elsewhere or that is copyrighted, the author needs to be authorized by the previous publisher or the copyright holder and/or indicate the reference source and copyrights. Please check and confirm whether the figures are original (i.e. generated de novo by the author(s) for this paper). If the picture is ‘original’, the author needs to add the following copyright information to the bottom right-hand side of the picture in PowerPoint (PPT): Copyright ©The Author(s) 2024.

Reply:
Dear Editor, we have provided the original image with the copyright information as per your comments

(2) Uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”.

Reply:
Dear Editor, we have changed the diagram note as per your request.

(3) Please provide the PubMed numbers (https://pubmed.ncbi.nlm.nih.gov/) and DOI citation numbers (https://doi.crossref.org/simpleTextQuery) to the reference list and list all authors of the references. If a reference has no PMID and DOI, please provide the source website address of this reference.

Reply:
Dear Editor, We have improved and supplemented the reference format as per your request.

(4) The author(s) must add a table/figure to the manuscript. There are no restrictions on the figures (color, B/W) and tables.

Reply:
Dear Editor, we have added figure to the manuscript as per your request

(5) The structure of Abstract does not meet the requirements. The abstract includes five parts: "BACKGROUND", "CASE SUMMARY", and "CONFUSION".

Reply:
Dear Editor, we have reformatted the abstract as per your request
(6) The “Case Presentation” section was not written according to the Guidelines for Manuscript Preparation.

Reply:
Dear Editor, we have rewritten the "Case Presentation" section in accordance with the Guidelines for Manuscript Preparation.

(7) Please provide the primary version (PDF) of the Consent for Treatment that has been signed by the patient(s) in the study/the first page of the patient(s)’ medical record, prepared in the official language of the authors’ country to the system.

Reply:
Dear Editor, we have provided the patient’s signed consent to treatment in the return document as per your request

(8) Please provide the filled conflict-of-interest disclosure form.

Reply:
Dear Editor, we have provided a conflict of interest disclosure form in the return at your request

(9) Please provide the CARE Checklist (2016).

Reply:
Dear Editor, We have provided the CARE Checklist (2016) in the return documentation as per your request.